Author's response to reviews

Title: The Dental Heath of primary school children living in fluoridated, pre-fluoridated and non-fluoridated communities in New South Wales, Australia

Authors:

Anthony S Blinkhorn Prof (anthony.blinkhorn@sydney.edu.au)
Pathik Methta Dr (dr.pathik.meththa@gmail.com)
Roy Byun (roy.byun@sswahs.nsw.gov.au)
George Johnson (george.johnson@sydney.edu.au)
Meredith Kay (meredith.kay@health.nsw.gov.au)
Peter Lewis (peter.lewis@health.nsw.gov.au)

Version: 2 Date: 20 May 2014

Author's response to reviews:

This manuscript is yet to be sent for review.